Author's response to reviews

Title: The motor development of orphaned children with and without HIV: Pilot exploration of foster care and residential placement

Authors:

Jennifer Jelsma (jennifer.jelsma@uct.ac.za)
Naila Davids (starphysio@yahoo.com)
Gillian Ferguson (gillian.ferguson@uct.ac.za)

Version: 4 Date: 10 January 2011

Author's response to reviews:

We have amended the on-line abstract to match the manuscript abstract.

We have deleted the section on what the manuscript adds and the biographical section. We have amended the headings from Introduction to Background and changed the heading of Conclusions and Recommendations to Conclusions in order to conform with the journal's style.

We have also acknowledged Prof Hadders-Algra and Prof Gabbard for their very useful reviews of the paper.